Author's response to reviews

Title: Evidence for an association of HLA-DRB1*15 and DRB1*09 with leprosy and the impact of DRB1*09 on disease onset in a Chinese Han population

Authors:

Fu-Ren Zhang (zhangfuren@hotmail.com)
Hong Liu (hongyue2519@hotmail.com)
Shu-Min Chen (chenshum@publicin.sd.cn)
Chang-Yuan Wang (wangchangyuan5@hotmail.com)
Chuan-Fu Zhu (sdhla0@yahoo.cn)
Lin Zhang (zhanghailin@yahoo.com.cn)
Tong-Sheng Chu (tongshengchu@163.com)
Dian-Chang Liu (liudianchang@yahoo.com.cn)
Xiao-Xiao Yan (yanxiaoxiao1979@163.com)
Jian-Jun Liu (liuj3@gis.a-star.edu.sg)

Version: 8 Date: 1 December 2009

Author's response to reviews:

Dear Editors:

Thank you very much for your appreciation of our manuscript. In the revised manuscript, we have improved some typographical mistakes.

We are looking forward to hearing a favorable response from you soon.

Yours sincerely,

Fu-Ren Zhang, MD, PhD

Shandong Provincial Institute of Dermatology and Venereology, 57 Jiyan Lu, Jinan 250022, Shandong Provience, P.R. China
Email: zhangfuren@hotmail.com
Tel: 86-531-87298801